IZA DP No. 355

Disentangling Treatment Effects of Polish Active Labor Market Policies: Evidence from Matched Samples

Jochen Kluve Hartmut Lehmann Christoph M. Schmidt

September 2001

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Disentangling Treatment Effects of Polish Active Labor Market Policies: Evidence from Matched Samples

Jochen Kluve

Alfred Weber Institute, University of Heidelberg and IZA, Bonn

Hartmut Lehmann

Heriot-Watt University Edinburgh, WDI University of Michigan, Ann Arbor, EERC Kiev and IZA, Bonn

Christoph M. Schmidt

Alfred Weber Institute, University of Heidelberg, CEPR London and IZA, Bonn

Discussion Paper No. 355 September 2001

IZA

P.O. Box 7240 D-53072 Bonn Germany

Tel.: +49-228-3894-0 Fax: +49-228-3894-210 Email: iza@iza.org

This Discussion Paper is issued within the framework of IZA's research areas *Labor Markets in Transition* and *Project Evaluation*. Any opinions expressed here are those of the author(s) and not those of the institute. Research disseminated by IZA may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent, nonprofit limited liability company (Gesellschaft mit beschränkter Haftung) supported by the Deutsche Post AG. The center is associated with the University of Bonn and offers a stimulating research environment through its research networks, research support, and visitors and doctoral programs. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public. The current research program deals with (1) mobility and flexibility of labor markets, (2) internationalization of labor markets and European integration, (3) the welfare state and labor markets, (4) labor markets in transition, (5) the future of work, (6) project evaluation and (7) general labor economics.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character.

IZA Discussion Paper No. 355 September 2001

ABSTRACT

Disentangling Treatment Effects of Polish Active Labor Market Policies: Evidence from Matched Samples *

This paper estimates causal effects of two Polish active labor market policies – Training and Intervention Works – on employment probabilities. Using data from the 18th wave of the Polish Labor Force Survey we discuss three stages of an appropriately designed matching procedure and demonstrate how the method succeeds in balancing relevant covariates. The validity of this approach is illustrated using the estimated propensity score as a summary measure of balance. We implement a conditional difference-in-differences estimator of treatment effects based on individual trinomial sequences of pre-treatment labor market status. Our findings suggest that Training raises employment probability, while Intervention Works seems to lead to a negative treatment effect for men. Furthermore, we find that appropriate subdivision of the matched sample for conditional treatment effect estimation can add considerable insight to the interpretation of results.

JEL Classification: C41, J68

Keywords: Active labor market policy, transition, exact matching, propensity score

Christoph M. Schmidt Department of Economics University of Heidelberg Grabengasse 14 69117 Heidelberg Germany Tel.: +49-6221-542955 Fax: +49-6221-543640 Email schmidt@uni-hd.de

^{*} The authors are grateful to Boris Augurzky, to participants of the first joint IZA/WDI conference on "Labor Markets in Transition Countries" in May 2000 in Bonn, and to seminar participants at the University of Heidelberg for helpful comments. Research Assistance by Markus Müller is gratefully acknowledged. Lehmann and Schmidt are grateful to the Volkswagen Foundation for financial support within the project "Analysis of Labour Market Adjustment in Poland and Estonia with Large Micro Data Sets". Kluve acknowledges financial support from the Cusanuswerk.

1. Introduction

The evaluation of active labor market policy (ALMP) in the transition countries of Central and Eastern Europe faces serious methodical obstacles. Most importantly, studies typically have to rely on nonexperimental data, a feature they share with most evaluation studies on measures of active labor market policy in OECD countries. In fact, nonexperimental settings are still predominant in any European country study, as large-scale – or any – experimental studies similar to those conducted in the US have remained highly uncommon.

Apart from this more general drawback early evaluation studies on transition countries frequently had to be based on yet inadequate data: certainly, first of all, local national statistics offices had to gather experiences in generating data sets. Moreover, as the urge to evaluate programs already emerged almost simultaneously with the introduction of the data sets and the introduction of the policy measures themselves, early studies could not exhaust any long-term data. And yet another distinct feature of policy evaluation in a transition country is the need to control for the – in early years after transition – quickly changing macro environment, in particular if one aims at estimation of individual treatment effects.

The transition countries of Central Europe display a U-shaped pattern of output over the first years of transition, showing an initial contraction in economic activity after the onset of reform followed by, in the Polish case, robust expansion (cf. Blanchard 1997). The effectiveness of ALMP measures depends – *ceteris paribus* – on the tightness of the labor market and, therefore, on the point on the U-curve where the economy is located. Evaluating the effects of ALMP measures administered over several years without controlling for the large moves along the U-curve observed in Central European transition countries would severely bias the results.

This study focuses on the evaluation of active labor market policy in Poland, with an emphasis on two major points. First, with regard to the implicit missing data problem in any nonexperimental evaluation study, we explore the potential of different matching procedures to achieve covariate balance, and we demonstrate how in our case exact matching methods may in an intuitively appealing way resolve the dilemma of constructing an adequate counterfactual. To this end we discuss three stages of a matching procedure that is meticulously adapted to the specific nature of the data. Our arguments are illustrated by comparing covariate balance and balance in estimated propensity scores – a summary measure of balance – across post-match samples.

Second, we discuss our evaluation results in detail, confirming earlier results on Polish ALMP (cf. Kluve, Lehmann and Schmidt 1999, Puhani 1998). We place particular emphasis on the necessity of considering subsets of the population of treatment units in the interpretation of results. We argue emphatically that a careful interpretation of results is as important as the devotion of effort to constructing an adequate comparison group, an idea that frequently seems to be overlooked in applied work. Specifically, we demonstrate that – even though an appropriate matching method does control for the relevant variables – once the comparison group is found, the analysis is not complete. Instead, pursuing the estimation of conditional treatment effects for appropriately defined subsamples may be useful to avoid otherwise misleading results.

The paper is organized as follows: Section 2 presents a brief description of the data and gives a short exposition of the evaluation problem, showing how matching on covariates and/or the propensity score can identify the treatment effect. Section 3 explains how our matched samples were constructed and to what extent the matching methods applied succeed in balancing observable covariates. Section 4 focuses on developing our matching estimator of treatment effects, on interpreting treatment effect estimates on covariates for interpretation purposes. Section 5 concludes.

2. Data and Methods

2.1 The Data

We employ data from the 18th wave of the Polish Labour Force Survey (PLFS) as of August 1996. The PLFS is a quarterly rotating panel introduced in May 1992. The distinct feature of the August 1996 wave is a supplementary questionnaire containing retrospective questions on individual labor market behavior. From these questions, individual labor market histories in quarterly structure have been constructed. The individual histories cover the 56-month-period from January 1992 to August 1996. Yet, the retrospective data required considerable preparatory work.

First, out of an initial number of 48,385 observations 11,102 individual labor market histories lacked any entry, and were omitted from the analysis. The vast majority

of these are individuals who were inactive in August of 1996. From the remaining data we had to exclude both treatment participants with too early (before January 1993) or too late (after November 1995) treatment spells since in our econometric approach we condition on pre-treatment histories spanning one year and look at post-treatment labor market outcomes averaged over three quarters. Incomplete spells containing too little information were also excluded from the analysis.

Our analysis focuses on individuals who experienced at least one spell of unemployment during the observation period. For both treated units and potential comparison units this ensures consideration of individuals potentially eligible for participation in ALMP measures offered by the employment offices. Since we focus on two distinct ALMP programs, Training and Intervention Works, the resulting samples of treatment participants for both measures and their potential comparisons are substantially smaller than the initial data set. We discuss sample composition in more detail in section 3.1.

Secondly, in order to be able to handle such rich data, we had to condense the information contained in individual labor market histories. Monthly entries entail, e.g., "employed", "unemployed", "receiving unemployment benefits", "maternal leave", etc. Furthermore, individual histories indicate whether and when an individual took part in an ALMP course. We compress the 30 possible monthly states occurring in the data into the three labor market states "employed" (henceforth denoted "1"), "unemployed" (denoted "2"), and "out-of-the-labor-force" (denoted "0"). Information on treatment participation is stored separately. Kluve et al. (1999) give a more detailed account of data transformation and adaptation. The resulting structure of individual spells for treatment and potential comparisons will be illustrated further in section 3.2.

In our estimation of individual treatment effects we consider two distinct measures of Polish ALMP: Training and Intervention Works¹. For more information on institutional details, on ALMP regulations and descriptions of courses we refer to earlier papers on the topic (Kluve et al. 1999, Puhani 1998, Góra and Schmidt 1998). For our purposes in this study it is mainly important to note the distinct nature of the two programs. Training is meant to enhance, or at least sustain, individual human capital during a period of unemployment. The Polish Training measure for the unemployed is

¹ A third measure of Polish ALMP, Public Works (=direct job creation in the public sector), has been left out in this study for the sake of brevity, and due to small sample sizes. Cf. also Kluve et al. (1999), Puhani (1998).

training off-the-job whose final aim is raising the unemployed person's probability of re-employment in a regular job.

Wage subsidy schemes like the Polish Intervention Works also have a human capital enhancing or -preserving aspect. However, the enhancement or preservation of a person's human capital takes place on-the-job. This human capital component of the program is thought to increase the chances of a participant to find regular, non-subsidized employment at the same firm or elsewhere after the end of the program. In addition, if there is asymmetric information about the productivity of potential employees, wage subsidy schemes are designed to facilitate temporary job matches that might translate into regular and lasting matches at the same firm once the subsidy ends. A crucial feature of ALMP regulation in the reported period, however, was that participation in Intervention Works was considered by the law like any other employment spell entitling individuals to a new round of benefit receipt, given the subsidized job lasted at least six months. Taking part in a Polish training measure for the unemployed did, on the other hand, not entitle a person to renewed benefit payments since this training was done off-the-job.

2.2 Matching as a substitute for randomization

Program evaluation aims at estimating causal effects, i.e. changes in the variable of interest that are due to treatment participation. The formal setting is cast into the statistical "potential outcome framework" for causal inference based on Neyman (1923 [1990], 1935), Fisher (1935) and Rubin (1974, 1977). Let us consider a population indexed by *i*, and let Y_{il} denote the variable of interest given individual *i* participated in a program, indicated by $D_i=1$. Likewise, let Y_{i0} denote the outcome if $D_i=0$, i.e. if individual *i* was not a participant, and define the single unit treatment effect as $D_i=Y_{il}$. However, outcomes Y_{il} and Y_{i0} are "potential" in that we can never observe both of them simultaneously for one individual. The parameter of interest in nonexperimental studies is the mean effect of treatment on the treated population:

(1)
$$\Delta|_{D=1} = E(\Delta_i | D_i = 1) = E(Y_{i1} | D_i = 1) - E(Y_{i0} | D_i = 1)$$

The equation shows the inherent missing data problem, as we cannot observe the nontreatment outcome Y_{i0} for treatment participants $D_i=1$. We thus have to rely on establishing a convincing substitute for $E(Y_{i0}/D_i=1)$ in equation (1) in order to identify the desired parameter.

In an experimental study randomization ensures that potential outcomes Y_{i1} and Y_{i0} are independent of treatment assignment D_i , i.e. $Y_{i1}, Y_{i0} \wedge D_i$. Hence, program participants and comparison group do not systematically differ from each other, yielding the expectation of Y_{i0} for the comparison group as a substitute for the expectation of Y_{i0} of the treated group. Thus,

(2a)
$$E(Y_{i0} | D_i = 1) = E(Y_{i0} | D_i = 0) = E(Y_i | D_i = 0),$$

where Y_i is the actually observed value of the outcome variable, i.e. $Y_i = D_i Y_{il} + (1-D_i) Y_{i0}$. Thus, randomization ensures identification of the desired parameter $D_{D=1}$ from equation (1). Randomization also implies an assumption referred to as stable-unit-treatment-value assumption (SUTVA, see e.g. Rubin 1980): Potential outcomes for each individual are not related to the treatment status of other individuals, i.e. $Y_{i0}, Y_{i1} \wedge D_j$ " i^1j .

Given a nonexperimental setting it appears appropriate to substitute for missing randomized-out controls by constructing a set of potential comparison units for whom we observe the same set of pre-treatment covariates X_i as for the treated units. The following proposition given in Rubin (1977) extends the above framework to nonexperimental studies:

If for each unit we observe a vector of covariates X_i , and $Y_{i0}, Y_{i1} \wedge D_i | X_i$ holds " *i*, then the population treatment effect for the treated $\mathbf{D}_{|D=1}$ is identified: it is equal to the treatment effect conditional on covariates and assignment to treatment $\mathbf{D}_{|D=1,X}$ averaged over the distribution $X|D_i=1$.

Such a construction of counterfactual outcomes can only be sensible if conditioning is on variables which itself are not the outcome of treatment participation. Post-treatment employment success is a case in point: by matching those individuals who are or are not successful, the effect of treatment will necessarily be derived to be zero. Similar conceptual reservations would hold for characteristics of post-treatment jobs such as industry or working hours. Consequently, conditional on observable covariates assignment to treatment can be considered as having been random, and unobservable characteristics possibly influencing treatment participation are ruled out. In fact, by this proposition comparing a program participant with a comparison individual displaying the same observable characteristics is like comparing the two in a randomized experiment. We thus merely need to estimate $E(Y_{i0} | X_i, D_i=0)$, so that

(2b)
$$E(Y_{i0} | D_i = 1) = E_X (E(Y_{i0} | X_i, D_i = 0) | D_i = 1),$$

identifying the mean effect of treatment on the treated of equation (1) for a nonexperimental setting: constructing the appropriate weighted average over conditional (on X) no-treatment outcomes mimics randomization by balancing all relevant covariates.

Ideally, in order to implement a procedure for estimating the conditional treatment effect $D_{D=1,X}$, we could simply match treated and comparison units on their covariate vector X_i . While exact matching on X_i achieves an exact balancing of attributes, it suffers from the fact that X_i might be of high dimension or contain continuously-distributed variables, so that some treated units might not find comparisons. To avoid the problem of matching on a high-dimensional X_i , the method of propensity score matching has been proposed by Rosenbaum and Rubin (1983). Define the propensity score as $p(X_i)=Pr(D_i=1/X_i)=E(D_i/X_i)$, i.e. the conditional probability of receiving treatment given a set of covariates. Then the conditional independence result from above extends to the propensity score: $Y_{i0}, Y_{i1} \wedge D_i/P(X_i)$.

The reduced dimension comes at a cost, however. The propensity score is not known and has to be estimated. Also, in samples of limited size, for some *i* and *j* it may occur that $p(X_i)=p(X_j)$ even if $X_i \, {}^{1}X_j$, resulting in imperfect balancing of the distributions of covariates. Thus, the small sample performance of propensity-score matching might be quite dismal. In fact, the literature indicates that the trade-off between exact matching and propensity score matching² is one of truly empirical nature: The decision for one approach or the other should depend heavily on the data, e.g. the number of

² In practice, matching algorithms are manifold, including e.g. exact matching, matching within calipers (fixed or flexible), minimum-distance matching, or optimal full matching minimizing total distance. For further reference see Gu and Rosenbaum (1993), Rosenbaum (1995), and Augurzky (2000).

observations, the dimension of X, time structure of variables, etc., and certainly it should depend on what the researcher believes (and justifies) to be the adequate *modus operandi* in each specific case.

Angrist and Hahn (1999) make this point forcefully by stating that existing theory provides little in the way of specific guidelines as how to choose between the two. On the one hand Hahn (1998) proves that exact matching is asymptotically efficient while propensity score matching is not, and concludes that asymptotic arguments would appear to offer no justification for anything other than full control for covariates. On the other hand Angrist and Hahn (1999) show that in some plausible scenarios estimators controlling for the propensity score can be more efficient than exact-matching estimators. The latter seems to be valid in particular when cell-sizes are small, the explanatory value of the covariates is low conditional on the propensity score, and/or the probability of treatment is far from ½.

Still, what counts in practice is how well balance is achieved, so that the researcher can indeed "compare the comparable" (Heckman et al. 1997). Any matching procedure allowing for any distance in either X or p(X) must be aware of that. And including a weak predictor of p(X) into the estimation might be more harmful than covariate matching on a reduced set of comparisons. Thus, both Dehejia and Wahba (1998) – based on an empirical study – and Augurzky and Schmidt (2000) – based on a simulation study – argue that it is more important to achieve balance of relevant covariates rather than painstakingly modeling the selection process.

3. Analyzing Matched Samples

3.1 Composition of Matched Samples

For each of the two measures under scrutiny – Training and Intervention Works – we start the construction of matched samples from an initial sample consisting of treated individuals and untreated potential comparison individuals, where every observation is required to have at least one spell of unemployment. From this starting point we subsequently impose stronger restrictions on X (i.e. enlarge the dimension of the matching criteria) step-by-step, in order to obtain three samples of matched treatment-comparison units for each of the two measures:

Sample A: A comparison unit is matched to a treated unit if his or her labor market history is observed without substantial gaps from a year before up to the beginning of treatment <u>and</u> from the end of treatment until 9 months later. None of the observed individual characteristics is used as a matching criterion.

Sample B: A comparison unit is matched to a treated unit if requirement (A) is met, and if he or she is identical in observable characteristics age, gender, education, marital status, and region.

Sample C: A comparison unit is matched to a treated unit if requirements (A) and (B) are met, and if he or she displays an identical 4-quarter (12-month) pretreatment labor market history at the exact same point in time as the treated unit.³

Samples (A) through (C) are constructed applying an exact-matching-within-calipers algorithm. For all three samples, if a treated individual finds any matching partner among the potential comparisons, this observation is retained. All algorithms allow for an oversampling procedure, i.e. a treated unit may be assigned more than one comparison unit. While we could have sharpened the matching criteria in a different order, this sequence reflects our conviction that *timing* is the pivotal aspect of comparison group construction in a transition economy.

The firmness in requirements (A) to (C) increases substantially. While under the weak precondition of Sample (A) no treated unit is lost in the matching process, and almost all potential comparisons are used, under requirement (C) some treated units do not find matching partners, and the number of matched comparison units is far smaller. Thus, algorithm (C) proceeds with replacement: some comparison units are matched to more than one treated individual. Samples (A) and (B) are constructed from potential comparison units with replacement, too, but here we use only the join of sets over matched comparison units.

Table 1 presents resulting sample sizes, as well as means of relevant variables. We observe that there is a reduction in the number of treated units who find matching

³ We consider 6 age categories, 3 education categories, gender, marital status, and 49 regions, resulting in 3528 different cells for sample (B). Including a 4-quarter sequence of a trinomial labor market outcome variable (cf. section 3.2) increases the number of cells to $3528*3^4=285,768$ cells for sample (C).

partners from (A) to (C) of almost one third for Training, and almost one quarter for Intervention Works. Due to matching-with-replacement, samples (C) contain comparison units matched to more than one treated unit. With less than one percent, the number is very low for Training, and with approximately one tenth it is also fairly low for Intervention Works. Table 1 also shows that Training participants on average are better educated, somewhat younger and more likely to be female than Intervention Works participants.

< Table 1 about here >

Throughout, we focus our attention on exact matching procedures. In sample (B), the number of matching variables is limited, and they are all categorical variables. Moreover, exact matching performs quite well: despite the substantial number of cells, approximately 9 out of 10 of treated units find a comparison unit. With regard to sample (C), our exact matching approach is a very practical device to account for the pre-treatment employment sequence. Further illustration is provided in the next sections.

3.2 Timing of interventions

In our preferred sample (C) we require treated and matched comparison units to display an identical pre-treatment history. To achieve comparability across samples (A) to (C), we impose the requirement on samples (A) and (B) that we observe any history at all in the year preceding treatment, although the precise information *what* history was experienced exactly is not used in matching. Moreover, to allow an assessment of posttreatment labor market performance, we require for treated units and all comparison samples that we observe a post-treatment sequence of labor force status variables in the nine months after treatment. In accordance with our preparatory data work, we condense the monthly information for treatment units to a sequence of three quarters of a multinomial outcome variable (0,1,2) denoting labor force status (out-of-the-labor force, employed, unemployed).

Correspondingly, for those comparison units eventually matched to the treated units, a comparable three-quarter post-treatment multinomial sequence of labor force status is computed as well, again starting at the exact point in time when the treatment spell of the corresponding treated unit ended. Our analysis thus incorporates individual treatment duration by conditioning on a complete (i.e. without major gaps) pretreatment labor market history being observed before month "start" and comparing labor force status outcomes after month "stop". Thus, treated units and matched comparison units are always being compared during the same period. Figure 1a illustrates this procedure for samples A and B, in which the timing structure is considered, but the contents of individually matched labor force status histories does not matter. Figure 1b proceeds to depict the case for inclusion of exact pre-treatment histories in matching for sample (C)

< Figures 1a,1b about here >

We thus take advantage of the specific nature of the data with monthly information on employment status for a 56-month period, considering the exact timing of "start" and "stop" of treatment – a feature that is neither common nor possible in many studies, even those focussing on duration data. Moreover, given the rapid upward moves of the Polish economy along the positive section of its U-shaped curve of output between 1992 and 1996, we can assume that labor market tightness has increased in Poland in the reported period. Hence, the fact that we are able to compare treated and comparison units individually at the same point of time seems particularly valuable.

There might be other ways to solve the crucial problem of finding the "starting point of treatment" for comparison units. In principle, one could first match on characteristics X or the propensity score conditioned on characteristics, p(X), and then directly impose requirements on comparable timing. A procedure following such a "partial balancing score" is for instance used by Lechner (2000). It seems more natural to us, however, to incorporate timing as a principal component of matching.

3.3 Covariate balance

In section 2.2 we have emphasized that balance in all relevant factors – observed as well as unobserved – is the principal objective in experiments, and in its observational counterpart, the matching approach. In this section we examine whether the particular matching procedures we applied here indeed succeed in balancing the distributions of pre-treatment covariates between treatment units and their comparisons. Figures 2 and 3 show the distributions of the two principal covariates age and region for treated and comparison units when matching is according to requirements (A) and the analyzed treatment is Intervention Works. By contrast to sample (A), samples (B) and (C) match on these individual characteristics. The figures illustrate by how much matching on the correct timing alone would miss out on balancing individual characteristics.

< Figure 2 about here >

Figure 2 shows that if not accounting for age, the young would be over-represented among the comparisons, and the mature (35-50, say) workers would be over-represented among the treated units.

< Figure 3 about here >

Figure 3 plots the frequency distribution for the 49 Polish voivodships. Including regional indicators among the matching covariates is firmly advocated by Heckman et al. (1997) in order to control for the local labor market. This is the more imperative in the Polish case, since local labor market conditions are quite heterogeneous in any typical transition country. The matching criteria for samples (B) and (C) achieve complete balance – besides oversampling of comparison units – in the distribution of voivodships for treated and comparison units, while sample (A) displays considerable imbalance. Thus, if regional information were left out of the matching algorithm, regional balance would not be assured.

With respect to further socio-demographic characteristics, 59.6% of Intervention Works participants are male, while there are only 47% men in comparison sample (A). Regarding the three education categories, the middle category comprises 63.6% of Intervention Works participants, and there is only one single individual out of the 275 treated (=0.36%) in the top category. Among comparison units in (A), 2.4% and 78.4% are in the top and middle categories, respectively. Table 1 shows that sample (B) and in particular sample (C) achieve balance in terms of sex and education.

3.4 Pre-Treatment Histories

The literature on program participation has always been concerned with the focal problem of controlling for observable characteristics, unobserved heterogeneity, and selection bias. Mainly affecting a difference-in-differences estimation approach, Ashenfelter (1978) pointed to a potentially serious limitation of this procedure when he observed a relative decline in pretreatment earnings for participants in subsidized

training programs. This empirical regularity has been called "Ashenfelter's dip" and has been confirmed by subsequent analyses of many other training and adult education programs (cf. Bassi 1983, Ashenfelter and Card 1985, LaLonde 1986, Heckman, LaLonde, and Smith 1999). For instance, Ashenfelter and Card (1985) apply a model that focuses on earnings changes as the determinants of participation. This line of thought was a natural consequence of Ashenfelter's discovery and resulted in analyses using earnings histories to eliminate differences between participants and nonparticipants⁴. Clearly, the fact whether the pre-program earnings dip is transitory or permanent determines what would have happened to participants had they not participated, and the validity of any estimation approach depends on the relationship between earnings in the post-program period and the determinants of program participation (Heckman and Smith 1999).

This rather established observation that it is earnings dynamics that drive program participation has lately been put into serious question by Heckman and Smith (1999), who argue that it is rather labor force dynamics that determine participation in an ALMP program. This point had implicitly been made before by Card and Sullivan (1988), who analyze training effects conditional on pre-program employment histories. Furthermore, Heckman and Smith (1999) argue for a distinction between employment dynamics – indicating whether an individual is employed or not – and labor force dynamics, incorporating also whether a nonemployed person is either unemployed or out-of-the-labor-force. Their conclusion is "that labor force dynamics, rather than earnings or employment dynamics, drive the participation process" (Heckman and Smith 1999). Therefore, we extend the "employment history setting" considered in Card and Sullivan (1988) to a "labor force status history setting" reflecting also movements in and out of inactivity.

We consider the 12-month labor market history of every single treated unit directly preceding the exact point in time – i.e. month – that the individual entered the program. As for the post-treatment outcomes, we condense the monthly information to a sequence of four quarters of a multinomial outcome variable (0,1,2) denoting labor force status (out-of-the-labor-force, employed, unemployed). For each treated unit in succession, the matching algorithm for sample (C) computes labor market histories for all potential comparison units at this point in time and matches those units who – in

⁴ Heckman and Smith (1999) attribute this emphasis also to the limited data available to "early analysts".

addition to the correspondence in the other covariates – display identical "pretreatment" histories. For illustration see Figure 1b.

Figures 4 and 5 draw the distributions of pre-treatment labor market histories for samples (A) and (B) for both Intervention Works (fig.4) and Training (fig.5). Representing a 12-month labor force status sequence with 4 quarterly realizations of a trinomial variable (0,1,2) yields 81 possible sequences ("0000" to "2222"). For the purpose of illustrating the balanced distributions – and only for that purpose – we classify these 81 sequences into 11 categories (see Appendix A), so that on the abscissa the low categories contain "inactive" sequences (mostly '0's), the middle categories comprise "unemployed" sequences ('2's), and the high categories represent "employed" sequences ('1's). Categories 1, 6, and 11 exclusively embody the straight sequences (i.e. "0000", "2222", and "1111", respectively).

< Figures 4 and 5 about here >

Thus, of the three peaks we observe in most of the graphs in figures 4 and 5, the left peak represents the area of "inactive" histories, because histories with a low order number contain many '0's. Accordingly, the peak in the middle expresses "unemployed" histories, and the peak to the right depicts "employed" histories. In terms of balancing of distributions, the picture is almost the same for figures 4 and 5. Both samples (A) and (B) display only limited accordance in pre-treatment histories for treated and comparison units. The figures also show that treatment individuals in Training are quite different from those in Intervention Works. For the Training participants, the fractions of "employed" and "unemployed" histories are quite close to each other, while in the Intervention Works sample we observe a far larger fraction of "unemployed" histories among the treated. Moreover, for both Training and Intervention Works the comparison samples (A) and (B) are too "successful" in that they contain too many "employed" sequences relative to "unemployed" sequences dominate.

3.5 Propensity score balance

The preceding sections were concerned with balance in selected individual characteristics. It is instructive to also provide a *summary measure of balance*, the propensity score. While the estimation of propensity scores is usually a principal step in

the construction of matched samples – with the hope that the resulting matched sample displays a balance in all relevant characteristics but no possibility to test this presumption – we can use our samples to directly analyze balance in the propensity score. Correspondingly, we predict post-match propensity scores for samples (A) and (B), based on estimates derived from sample (A). We follow a probit specification with interaction terms between some of the covariates,

(3)
$$P(D=1|X) = \Phi(a_0 + a_1X + a_2X \otimes X)$$

where F denotes the cumulative normal density function, X is the vector of covariates, and $X \otimes X$ indicates all relevant interactions across covariates. Regressors comprise indicator variables capturing age, education, gender, and region. Moreover, corresponding to the condensation of pre-treatment labor market histories into 11 distinct "types" in section 3.4, there are 10 indicators of pre-treatment history among the regressors. Finally we interacted age, gender and education in a saturated fashion.

This model is estimated using the treatment units (yielding the value "1") and comparison sample (A) providing the "0" observations. Note that we observe both the individual characteristics and the pre-treatment histories also with comparison sample (A), although this information is utilized only in the construction of comparison samples (B) and (C), respectively. The resulting coefficients are employed to predict propensity scores in samples (A) and (B). Figures 6 and 7 document the distribution of propensity scores in these comparison samples – relative to the corresponding distribution among treatment units – for the two measures under study.

< Figures 6 and 7 about here >

Note that the density for treated units is not scaled relative to the number of observations in the comparison pool, so that the figure depicts the distribution of scores rather than the proportion of treated units to comparison units. In both figures 6 and 7 the comparison units gather at the low end of the estimated score. Whereas for Intervention Works treated units are distributed rather evenly, with the peak to the low end and then slightly declining towards the upper tail, the majority of treated units for Training also displays relatively low scores, with an overall distribution quite close to

that of comparison units. We find relatively little change in balance from (A) to (B) for both Training and Intervention Works. For Training the distributions are rather balanced – for Intervention Works, however, the substantial imbalance in pre-treatment histories clearly finds expression in the score distributions for (A) and (B) that do not yet control for this imbalance.

4. Empirical results

4.1 Distributions of outcomes

To illustrate the substantial heterogeneity of labor market outcomes following Intervention Works and Training, Figures 8 and 9 plot distributions for the post-treatment employment success for treatment units and the comparisons in samples (A) to (C). There are 27 possible labor market status sequences capturing employment performance in the three quarters succeeding treatment (cf. also Figures 1a, 1b). Similar to our presentation of pre-treatment labor market histories, we classify these 27 possible sequences of 3 quarterly realizations of a trinomial variable into 9 categories for illustration purposes. This categorization is outlined in Appendix A. Once more, low categories contain "inactive" sequences (category 1="000"), middle categories include "unemployed" sequences (category 5="222"), and high categories comprise "employed" histories (category 9="111"). Accordingly, in the graphs the left peak depicts "inactive" sequences, the middle peak "unemployed" sequences, and the right peak represents "employed" histories.

< Figure 8 about here >

Looking at the Intervention Works samples in Figure 8, we find that in all samples the "unemployed" sequences are clearly predominant for the treated units. At the same time, comparison units display rather successful labor market histories in samples (A) and (B). For our preferred comparison sample (C) this picture changes considerably, and a larger fraction of comparison units also displays "unemployed" histories. However, the comparison group still fares visibly better than the program participants. Attributing the most reliable results to sample (C), we would conclude that during the 9 months directly succeeding participation in Intervention Works the treated units on average were

marginally – possibly insignificantly – less successful in finding employment than the comparison units.

< Figure 9 about here >

For the Training samples shown in Figure 9 we find slightly different results. Similar to what we have seen for the pre-treatment sequences of these samples (Figure 5), the "employed" and "unemployed" peaks have more or less the same height also for the post-treatment sequence. But while for samples (A) and (B) the "employed" peak is higher for comparison units than for treated units, and the "unemployed" peak is higher for treated units than for comparison units, this relation switches for sample (C). In (C) treated units display on average a slightly more successful post-treatment labor market sequence than corresponding comparisons. We would thus attribute a slightly – possibly insignificant – positive treatment effect to Training.

Taken together, Figures 8 and 9 display three important patterns. First, moving from (A) to (C) we do not observe much variation in the distributions for treated units. Thus, the fact that we lose some treated units while increasing matching requirements does not seem to play an important role. Second, without conditioning on pre-treatment labor market histories the comparison samples apparently contain too many "successful" individuals – a pattern which we already observed for pre-treatment labor force status sequences in Figures 4 and 5. For samples (A) and (B) this would result in a far too negative estimate of treatment effects. Third, across comparison units and treated units we observe clearly more "successful" outcomes for Training than for Intervention Works. This is not surprising, as we noticed a similar relation for pre-treatment labor market history distributions (Fig. 4 and 5).

In Figures 10 and 11 we address the idea that participation in Intervention Works might primarily be a vehicle to renew eligibility for unemployment benefits. Recall that according to Polish ALMP regulations Intervention Works renews benefit receipt eligibility, whereas Training does not. Figures 10 and 11 perform a simple before-after comparison of the variable "unemployment benefit receipt" for both ALMP measures, and for men and women separately. The top panel of each figure indicates benefit receipt in at least two of the three months directly *preceding* treatment. The middle panel shows benefit receipt in at least two of the three in at least two of the three months directly succeeding treatment. The bottom panel plots benefit receipt in at least two months of each of the

three quarters succeeding treatment, i.e. at least 6 out of 9 months. We focus on sample (C) for both measures.

< Figure 10 about here >

Figure 10 shows for Intervention Works that a substantial fraction of both treated and comparison units received pre-treatment benefits, although benefits do seem to play a more important role for treated units. This pattern is more pronounced for men. In the middle and bottom panel this situation aggravates substantially. While both short-term and medium-term benefit receipt played a minor role for comparison units, we observe that approximately 60% of the treated males received unemployment benefits in the quarter directly following treatment, and that more than half of the treated males received benefits during the whole 9-month post-treatment period. For females, this pattern is not quite as severe, but still post-treatment benefit receipt plays a major role for Intervention Works participants.

< Figure 11 about here >

The situation for the Training sample is quite different. As Figure 11 shows, unemployment benefits do play some role for both treated and comparison units during the one quarter directly before and after participation, at least for the males. However, in the medium run this effect diminishes, and only very few observations in the treatment and comparison group display benefit receipt for the whole 9-month period following treatment. This pattern is even less pronounced for women than for men.

As a result, figures 8 through 11 indicate that individuals involved in Training measures seem to be generally more successful before and after the treatment than those participating in Intervention Works. However, these patterns are difficult to reconcile on the basis of a more favorable impact of Training. Rather, this simple evidence suggests that substantial benefit churning seems to take place in the case of Intervention Works, but not in the case of Training.

4.2 Treatment effect estimation

Our aim is to identify treatment effects of two different measures of Polish active labor market policy, Intervention Works and Training, which we consider separately in the empirical analysis. For purposes of the formal exposition of our estimation approach we consider a single generic intervention. Furthermore, we explicitly require that treated units be matched with comparison units from the identical set of observed pre-treatment and post-treatment months. Any reference to the time period is therefore omitted from the formal exposition as well.

In addition to the terminology introduced in section 2, let N_I denote the number of treated units, with indices $i \ \hat{I} I_I$, and N_0 the number of potential comparison units, with indices $i \ \hat{I} I_0$. Potential labor market outcomes in post-treatment quarter q (q = 1, 2, 3) are denoted by Y_{qi}^1 , if individual *i* received treatment, and by Y_{qi}^0 , if individual *i* did not receive treatment. Outcomes are defined as multinomials with three possible realizations ('0'=out-of-the-labor-force, '1'=employed, '2'=unemployed), extending the formulations of Card and Sullivan (1988) from a binomial to a trinomial setting.

We can only observe one of the two potential outcomes Y_{qi}^1 and Y_{qi}^0 for a given individual. This actual outcome is denoted by Y_{qi} . The objective is then to formally construct an estimator of the mean of the unobservable counterfactual outcome $E(Y_{qi}^0/D_i=1)$. Following the quarterly sequence of labor market outcomes might be too detailed, though, for a direct economic interpretation of results. Thus, to condense the available information further, the post-intervention labor market success of each individual *i* is summarized by the individual's average employment rate over the three quarters following the intervention. Using indicator function $\mathbf{1}(.)$, these employment rate outcomes are $\frac{1}{3}\sum_{q}\mathbf{1}(Y_{qi}=1)$.⁵ Observed outcomes for individual *i* can then be written as

(4)
$$\frac{1}{3}\sum_{q} \mathbf{1}(Y_{qi} = 1) = \frac{1}{3}(D_i \sum_{q} \mathbf{1}(Y_{qi}^1 = 1) + (1 - D_i)\sum_{q} \mathbf{1}(Y_{qi}^0 = 1))$$

⁵ Kluve et al. (1999) extend this setting to considering both employment and unemployment rates, so that corresponding outcomes would be $\frac{1}{3}\sum_{q} \mathbf{1}(Y_{qi} = w)$, where $w \hat{\mathbf{I}}$ {1,2}. Comparing employment and unemployment rate treatment effects shows for instance that exits to inactivity play a much larger role for women than for men. Moreover, Kluve et al. (1999) also consider the medium run, i.e. 6 post-treatment quarters, while we focus on the short-term case here. The extension to any number of post-treatment periods is straightforward.

and the impact of the intervention on the average labor market status of individual i can be expressed as

(5)
$$\Delta_i = \frac{1}{3} \left(\sum_q \mathbf{1}(Y_{qi}^1 = 1) - \sum_q \mathbf{1}(Y_{qi}^0 = 1) \right)$$

for average employment rates. The parameters of interest in our evaluation analysis are weighted population averages over these individual treatment effects, the mean effect of treatment on the treated for types of individuals characterized simultaneously by specific sets of characteristics X; and labor market histories before treatment h_i ,

(6)
$$E(\Delta_i | X_i, h_i, D_i = 1) = E(\frac{1}{3}(\sum_q \mathbf{1}(Y_{qi}^1 = 1) - \sum_q \mathbf{1}(Y_{qi}^0 = 1)) | X_i, h_i, D_i = 1)$$

The less inclusive the chosen set of characteristics conditioned upon – i.e. the more specific characteristics are included in X – the larger is the population of treated individuals over which the conditional mean is taken. As laid out above, previous labor market histories h_i are captured by the sequence of labor market states in the four quarters preceding the intervention.

Our approach to combine the population averages of the treatment effects for individuals in a given history-specific "cell" – characterized by demographic and other characteristics, in particular labor market history – gives us considerable flexibility in addressing the economic interpretation of results. The standard approach to evaluation would be to consider the distinction of type-history cells primarily as a device to achieve comparability of treatment and comparison units (see below). The ultimate interest there typically lies in the average treatment effects over the joint support of *X* and *h* given D=1,

(7)
$$M = \sum_{s} w_{s} E(\Delta \mid s, D = 1),$$

with *s* indicating any possible combination of *X* and *h*, and w_s representing the corresponding relative frequency in the treatment sample. By contrast to this standard approach, in what follows we will consider appropriate subsets of this joint support.

How does our particular observational approach – matching – facilitate the estimation of these parameters of interest? In randomized experiments the counterfactual expected values under no intervention can simply be estimated for intervention recipients by the mean values of the outcome for randomized-out would-be recipients. As we have shown in section 2, matching methods can recover the desired counterfactual for a nonexperimental comparison group: Within each matched set of individuals, one can estimate the treatment impact on individual *i* by the difference over sample means, and one can construct an estimate of the overall impact by forming a weighted average over these individual estimates.

Matching estimators thereby approximate the virtues of randomization mainly by balancing the distribution of observed attributes across treatment and comparison groups, both by ensuring a common region of support for individuals in the intervention sample and their matched comparisons and by re-weighting the distribution over the common region of support. The central identification assumption is that of mean independence of the labor market status Y_{qi}^0 and of the treatment indicator D_i , given individual observable characteristics. In our specific application these conditioning characteristics are the demographic and regional variables X_i and the pre-treatment history h_i , i.e. from equation (2) in our case,

(8)
$$E(\mathbf{1}(Y_{qi}^{0}=1)|X_{i},h_{i},D_{i}=1) = E(\mathbf{1}(Y_{qi}^{0}=1)|X_{i},h_{i},D_{i}=0)$$

Thus, by conditioning on previous labor market history we exploit the longitudinal nature of our data.

In a standard difference-in-differences approach pre-treatment and posttreatment outcomes are typically treated symmetrically; the identifying assumption is that the change in outcomes that treated individuals would have experienced had they not received treatment, would have been the same change – on average – that untreated individuals experience during the same period. This assumption accounts for the phenomenon that treatment units typically experience lower pre-treatment outcomes, even though they might be otherwise identical to comparison units. It does not lend itself naturally to the analysis of categorical outcome variables, though. In this context, a natural generalization of the difference-in-differences idea is to condition on the specific realization of the outcome variable in the pre-treatment period, as we do here. This is possible, since due to the categorical nature of the outcome the conditioning remains tractable. Card and Sullivan (1988) and Heckman et al. (1997) advocate such difference-in-differences approaches (cf. also Schmidt 1999).

Our matching estimator is one of oversampling exact covariate matching within calipers, allowing for matching-with-replacement. Our particular attention to pretreatment labor market histories implements this idea of a generalized difference-indifferences juxtaposition between treated units and comparison units. Due to the relevance of the previous history for subsequent labor market success – state dependence is one of the issues most discussed in the labor literature – we also emphasize this variable in the construction of the estimates. Specifically, for any treatment history h for which at least one match could be found, we estimate the impact of the intervention by

(9)
$$\hat{M}_{h} = \frac{1}{N_{1h}} \sum_{i \in I_{1h}} \left[\frac{1}{3} \sum_{q} \mathbf{1}(Y_{qi}^{1}=1) - \sum_{j \in I_{0h} \mid X_{j} \in C(X_{i})} \frac{1}{n_{i0}} (\frac{1}{3} \sum_{q} \mathbf{1}(Y_{qj}^{0}=1)) \right] ,$$

where N_{Ih} is the number of individuals with history *h* who receive the intervention $(N_1 = \sum_h N_{1h}), I_{Ih}$ is the set of indices for these individuals, $C(X_i)$ defines the caliper for individual *i*'s characteristics X_i , and n_{i0} is the number of comparisons with history *h* who are falling within this caliper, with the set of indices for comparison-individuals with history *h* being I_{0h} . The standard error of the estimated treatment effect is then constructed as a function of the underlying multinomial probabilities. This procedure is outlined in Appendix B.

The overall effect of the intervention is estimated in a last step by calculating a weighted average over the history-specific intervention effects,

(10)
$$\hat{M} = \sum_{h} \left[\frac{N_{1h}}{\sum_{h} N_{1h}} \hat{M}_{h} \right]$$

using the treated units' sample fractions as weights. The variance is derived as the corresponding weighted average of the history-specific variances.

4.3 Treatment effect results

In this section we analyze the treatment effect estimates which we obtain by applying the estimator developed in the previous section. Table 2 presents average treatment effects on the post-intervention employment rate for Intervention Works sample (C). The structure of the table shows how the total treatment effect (-.126) is being calculated by computing history-specific effects first. As explained above, for each treated unit, if he or she has more than one matched comparison unit, the comparison units' employment rates are averaged and handled as if they were the employment rate of only a single unit. The total effect is the weighted average of the history-specific effects using the treated units' sample fractions as weights.

< Table 2 about here >

Besides treatment effect calculation Table 2 shows which labor market state sequences occurred in the data, thus picking up the theme of figure 4. We observe the same predominance of "unemployed" histories which we already noticed in the figure. The total treatment effect casts a rather negative picture on the Intervention Works program, suggesting that participation tends to lower post-treatment employment prospects. In principle, this finding would conclude our analysis: we have described the nonexperimental context of the study, we have shown by what means we overcome the problem of constructing the desired counterfactual, and we have applied the appropriate estimation methods in order to obtain credible treatment effect estimates. As far as the data permit, the causal effect of Intervention Works participation is identified. Or is it?

In fact, looking at Table 3 we find that there may be more to it. First, we report treatment effect estimates for comparison samples (A) and (B) obtained by taking sample averages over the average employment rate in the three post-treatment quarters. The estimates are far more negative than the one obtained using sample (C), clearly reflecting the over-representation of "successful" labor force status sequences in the respective comparison samples (cf. Fig. 4 and 8). Furthermore, in accordance with our discussion of expression (7), in Table 3 we subdivide the matched Intervention Works comparison sample (C) with respect to various covariates, and we compare the conditional treatment effect for the subsample to the full sample estimate. Even a simple subdivision by gender reveals an interesting finding: The significantly negative full sample effect consists of a – more or less – zero treatment effect for women and a

considerably larger negative effect for men. On the other hand, a subdivision by date of program entry that parts the observation period into two halves does not reveal any apparent influence of changes in the macroeconomic environment.

< Table 3 about here >

The next step is to further refine cells and to classify the sample by both gender and date of program entry. These subsamples indicate that post-treatment employment prospects for male Intervention Works participants were quite unfavorable in the second period after July 1994, but particularly severe during the first period until June 1994. For women the time period distinction leads to the opposite result, but both the positive effect of the first half and the negative effect of the second half are small and insignificant. This also points to the fact that, as we increase the number of subdivisions, subsample sizes decrease and standard errors increase.

Classification by labor market history allows us to look at the two major labor force status sequences that drive the peaks from Figures 4 and 5. For "employed" (1111) histories subsample sizes are rather small and the effects not well defined. For the subsample of "unemployed" (2222) histories, which entails almost 80% of total treated and comparison units, we find a significantly negative treatment effect close to the full sample effect. This is certainly no surprise, as the estimate of the full sample effect is dominated by the "2222" subsample effect. If we further classify by labor market history and gender, treatment effects for the "1111" subsample remain insignificant for both men and women, while the "2222" subsample displays the same substantial male/female difference in the treatment effect that we have seen for the full sample.

Table 4 reports the same comparison between samples and various subdivisions for Training. Both treatment effect estimates from comparison samples (A) and (B) suggest an insignificantly negative effect of Training participation, while the estimate obtained from sample (C) indicates that Training raises the individual employment probability by 13.8%. This sudden switch of signs is in line with our observations drawn from Figure 9. Further looking at comparison sample (C), we conclude that in the case of Training a classification by gender does not seem to add any insights to the interpretation: Treatment effects for men and women are almost identical. While a categorization by gender and date of program entry shows contradictory results (upward for men, downward for women from one period to the other), the number of observations per subsample is in fact too small to draw any firm conclusions. Looking at a classification by labor market history, once more we find the "peaks" from Figure 5, indicating here that the share of "1111" sequences is almost as large as the that of "2222" sequences. Again, subsample sizes are quite small for interpretation purposes.

< Table 4 about here >

From these calculations results the observation that an appropriate subdivision of a matched sample can substantially contribute to disentangling and identifying heterogeneous treatment effects. In particular, the example of a simple classification by gender for the Intervention Works sample is striking: The overall negative effect is almost exclusively due to the dismal post-treatment labor market performance of male participants. Thus, while the recognition of the principal idea that treatment effects are heterogeneous across the population has led to the development of sophisticated econometric methods for constructing convincing counterfactuals, it is easy to forget the necessity to stratify the sample appropriately in order to interpret the results in economically meaningful terms. Thus, controlling for observable characteristics in establishing the statistical model does not seem to be sufficient – it appears to be good advice to re-consider the same observable characteristics (which we already controlled for) when analyzing the empirical results. This recommendation seems imperative if one wants to assess for example targeting issues: bad targeting of programs is often claimed to be one reason for disappointing treatment effects. In our particular application, Intervention Works has been uncovered as an extremely disappointing measure in the case of men - a result that would have remained hidden, had we not pursued an appropriate sample split.

Of course, these negative treatment effects could be explained by other factors than poor targeting. Stigma is often given as a reason why participants of an employment program like Intervention Works perform worse in the labor market than non-participants.⁶ Prospective employers identify participants as "low productivity workers" and are not willing to accept them into regular jobs. Another explanation, which might have particular merit in the Polish case, is benefit churning. Workers with long unemployment spells who have difficulty finding regular employment are

⁶ A large part of the intervention works jobs are actually in the public domain, i.e. we can also think of this scheme as a public employment program.

identified by labor bureau officials and might only be chosen for participation in an employment scheme so that they re-qualify for another round of benefit payment.

While the presented evidence cannot pinpoint precisely the cause underlying the poor labor market performance of males participating in Intervention Works, stigmatization seems to be the least likely cause. For if participation in the scheme was a bad signal to prospective employers, it is not clear why this would not be the case for female participants. It may be that those males – males are for the most part heads of households – are targeted by labor bureau officials who have especially poor prospects for regular employment. Once the publicly subsidized job comes to an end, so officials might reason, they at least qualify for another round of unemployment benefits, if they cannot find regular employment elsewhere or if their subsidized job is not transformed into a regular job. It is probably not a mere coincidence that the large majority of Intervention Works jobs lasts six months, the length of time one needs to work within the year preceding benefit receipt in order to qualify for unemployment benefits.

However, more work is needed to determine firmly the factor(s) that drive the poor labor market performance of males after their participation in Intervention Works comes to an end. For example, the fate of female participants after the end of the subsidized job needs to be more thoroughly analyzed. Specifically, one needs to ask whether female participants are more likely to be kept on by employers or whether they find regular jobs elsewhere more readily than men because their characteristics are better than those of men, i.e. because the targeting criteria are different for men and women. It could also be that women who participate in Intervention Works are selected into jobs that are more conducive to prolonged job matches because demand in these jobs is strong (e.g. nursing jobs).

In addition to displaying the treatment effects by sample and subdivision, Table 5 presents treatment effect estimates for comparison samples (C) obtained from a "counterfactual experiment". The first line reports the factual Intervention Works treatment effect estimate computed as shown in Table 2. This estimate tries to answer the question: "How much did Intervention Works participants benefit from participating in Intervention Works?" The second line reports a "counterfactual" Intervention Works treatment effect for Training participants, i.e. it tries to answer the question: "How much would Training participants have benefited, if they had participated in Intervention Works sample using the fraction of the treated units in the Training sample as weights. Looking

at Table 2 this is the same as if for each history the second column contained the corresponding number of observations from the Training sample. Apparently, this reweighting by labor market history implicitly assumes that there are no relevant changes in other elements of X.

< Table 5 about here >

The estimate in the second line of Table 5 shows that, while the Intervention Works effect on Training participants still displays a negative sign, the effect is insignificant, so that Training participants participating in Intervention Works would have done better than Intervention Works participants themselves. Looking at the effects of Training on Training participants and Intervention Works participants, respectively, we find the counterpart to this result: Intervention Works participants participants and much from the treatment as Training participants themselves. Thus, persons with better observable and unobservable characteristics seem to have been targeted for the Training program.

The last two lines in Table 5 report differential treatment effects of Intervention Works vs. Training. The estimates represent the difference between the difference of treated and comparison units in Intervention Works (second to last column, Table 2) and the difference of treated and comparison units in Training. Once more, differences are taken history-wise and weighted using either Intervention Works participants or Training participants sample weights. Both estimates clearly show that Training is the superior ALMP to Intervention Works.

The methodology used in our paper allows us to evaluate ALMP at the individual level. It thus tells us that those persons participating in Polish Training programs have better employment prospects than they would have had had they not participated and also that they have better employment prospects than those who take part in Intervention Works. The methodology does not address the issue whether Training improves the overall performance of the labor market, i.e., for example, whether it lowers the aggregate unemployment rate. Even if Training is beneficial at the individual level, substitution effects - Training participants just "jump the queue" of those in line for regular jobs - could neutralize its impact at the aggregate level. On the other hand, the finding that a program is not even effective at the individual level, like the Polish Intervention Works scheme, helps us to focus attention on targeting issues

and/or wrong incentive structures that distort the behavior of labor bureau officials and of the unemployed.

5. Conclusion

In this paper we have analyzed treatment effects of two Polish measures of active labor market policy: Training and Intervention Works. The analysis was based on matched samples to overcome the inherent evaluation problem of constructing a credible counterfactual in a nonexperimental setting. We have seen how matching methods can solve this problem by balancing distributions of relevant covariates. Matching methods can be based on exact-covariate-matching, propensity score matching, or a combination of both (partial score). We have argued that on both theoretical and above all empirical grounds the decision for one approach or the other depends heavily on the data.

We have illustrated our own approach to the data by the construction of three different comparison samples using exact-matching-within-calipers, imposing increasingly stricter preconditions. Figures 1 to 5 have depicted how strong requirements, i.e. a more detailed match on observable characteristics substantially improve the balancing of covariates, and thus the quality of the match. As long as sample sizes do not decrease considerably, such a procedure appears promising. We have illustrated the balancing property of our exact matching approach using the estimated propensity score as a summary measure of balance.

The estimation of the treatment effect is based on a history-specific generalized difference-in-differences estimator. Our estimates suggest that, while Training seems to clearly enhance individual employment prospects, Intervention Works participants fare substantially worse than their comparisons. This is in line with previous findings (cf. Kluve, Lehmann and Schmidt 1999, Puhani 1998). However, we do point to the fact that appropriate subdivision of the matched sample can add considerable insight to the interpretation of results. In our study, for instance, we find that the overall negative treatment effect of Intervention Works is almost exclusively due to the dismal employment performance of male participants, while women do neither gain nor lose anything by participating. From an empirical point of view, we thus doubt that controlling for covariates in constructing the counterfactual is sufficient to account for the heterogeneity of treatment effects – appropriate subdivision of the matched sample may often add clarity to the economic interpretation.

27

References

- Angrist, Joshua D., and Jinyong Hahn (1999), "When to Control for Covariates? Panel-Asymptotic Results for Estimates of Treatment Effects", *NBER Technical Working Paper* 241, Cambridge, MA.
- Ashenfelter, Orley (1978) "Estimating the Effect of Training Programs on Earnings", *Review of Economics and Statistics* 60, 47-57.
- Ashenfelter, Orley, and David Card (1985) "Using the Longitudinal Structure of Earnings to Estimate the Effect of Training Programs", *Review of Economics and Statistics* 67, 648-660.
- Augurzky, Boris (2000), "Optimal Full Matching", Dept. of Economics Discussion Paper 310, University of Heidelberg.
- Augurzky, Boris, and Christoph M. Schmidt (2000), "The Propensity Score: A Means to an End", *Dept. of Economics Discussion Paper* 334, University of Heidelberg.
- Bassi, Lauri J. (1983) "The Effect of CETA on the Post-Program Earnings of Participants", *The Journal of Human Resources* 18, 539-556.
- Blanchard, Olivier (1997), *The Economics of Post-Communist Transition*, Oxford: Clarendon Press.
- Card, David, and Daniel Sullivan (1988), "Measuring the Effect of Subsidized Training Programs on Movements In and Out of Employment", *Econometrica* 56, 497-530.
- Dehejia, Rajeev H., and Sadek Wahba (1998), "Propensity Score Matching Methods for Non-Experimental Causal Studies", *NBER Working Paper* 6829.
- Fisher, Ronald A. (1935), *The Design of Experiments*, Edinburgh: Oliver & Boyd.
- Góra, Marek and Christoph M. Schmidt (1998), "Long-term unemployment, unemployment benefits and social assistance: The Polish experience", *Empirical Economics* 23, 55-85.
- Gu, Xing Sam, and Paul R. Rosenbaum (1993), "Comparison of Multivariate Matching Methods: Structures, Distances, and Algorithms", *Journal of Computational and Graphical Statistics* 2, 405-420.
- Hahn, Jinyong (1998), ""On the Role of the Propensity Score in the Efficient Semiparametric Estimation of Average Treatment Effects", *Econometrica* 66, 315-332.
- Heckman, James J., Hidehiko Ishimura and Petra E. Todd (1997), "Matching as an Econometric Evaluation Estimator: Evidence from Evaluating a Job Training Programme", *Review of Economic Studies* 64, 605-654.

- Heckman, James J., Robert J. LaLonde, and Jeffrey A. Smith (1999), "The Economics and Econometrics of Active Labor Market Programs", in: Ashenfelter, Orley and David Card (eds.): *Handbook of Labor Economics*, vol. III, Amsterdam et al.: North-Holland.
- Heckman, James J., and Jeffrey A. Smith (1999), "The Pre-programme Earnings Dip and the Determinants of Participation in a Social Programme: Implications for Simple Programme Evaluation Strategies", *The Economic Journal* 109, 313-348.
- Kluve, Jochen, Hartmut Lehmann, and Christoph M. Schmidt (1999), "Active Labor Market Policies in Poland: Human Capital Enhancement, Stigmatization, or Benefit Churning?", *Journal of Comparative Economics* 27, 61-89.
- LaLonde, Robert J. (1986), "Evaluating the econometric evaluations of training programs with experimental data", *American Economic Review* 76, 604-620.
- Lechner, Michael (2000), "An Evaluation of Public-Sector-Sponsored Continuous Vocational Training Programs in East Germany", *The Journal of Human Resources* 35, 347-375.
- Neyman, Jerzy (1923 [1990]), "On the Application of Probability Theory to Agricultural Experiments. Essay on Principles. Section 9.", translated and edited by D.M. Sabrowska and T.P. Speed from the Polish original, which appeared in Roczniki Nauk Rolniczych Tom X (1923), 1-51 (Annals of Agriculture), *Statistical Science* 5, 465-472.
- Neyman, Jerzy (1935), with co-operation by K. Iwaszkiewicz, and S. Kolodziejczyk, "Statistical Problems in Agricultural Experimentation", (with discussion), Supplement to the Journal of the Royal Statistical Society 2, 107-180.
- Puhani, Patrick A. (1998), "Advantage through Training? A Microeconometric Evaluation of the Employment Effects of Active Labour Market Programmes in Poland", ZEW Disc. Paper 98-25, Mannheim.
- Rosenbaum, Paul R. (1995), 'Observational Studies'', New York: Springer Series in Statistics.
- Rosenbaum, Paul R., and Donald B. Rubin (1983), "The Central Role of the Propensity Score in Observational Studies for Causal Effects", *Biometrika* 70, 41-55.
- Roy, Andrew D. (1951), "Some Thoughts on the Distribution of Earnings", Oxford Economic Papers 3, 135-146.
- Rubin, Donald B. (1974), "Estimating Causal Effects of Treatments in Randomized and Nonrandomized Studies", *Journal of Educational Psychology* 66, 688-701.
- Rubin, Donald B. (1977), "Assignment to Treatment Group on the Basis of a Covariate", *Journal of Educational Statistics* 2, 1-26.

- Rubin, Donald B. (1980), Comment on "Randomization Analysis of Experimental Data: The Fisher Randomization Test" by D. Basu, *Journal of the American Statistical Association* 75, 591-593.
- Schmidt, Christoph M. (1999), "Knowing What Works The Case for Rigorous Program Evaluation", *IZA Discussion Paper* 77, Bonn.

Appendix A. Categorizing labor market status sequences

Pre-treatment

Category	1	2	3	4	5	6	7	8	9	10	11
Histories	0000	0001	0012	0022	2201	2222	2220	2211	1102	1110	1111
		0010	0102	0202	2021		2202	2121	1012	1101	
		0100	1002	2002	0221		2022	1221	0112	1011	
		1000	0120	0220	2210		0222	2112	1120	0111	
		0002	1020	2020	2012		2221	1212	1021	1112	
		0020	0021	2200	0212		2212	1122	0121	1121	
		0200	1200		2120		2122		1210	1211	
		2000	0201		2102		1222		1201	2111	
			0210		0122				0211		
			2100		1220				2110		
			2010		1202				2101		
			2001		1022				2011		
			0110						0011		
			1010						0101		
			1100						1001		

Post-treatment

Category	1	2	3	4	5	6	7	8	9
Histories	000	001 010 100 002 020 200	210 120 102	220 202 022	222	221 212 122	012 021 201	110 101 011 112 121 211	111

Appendix B. Calculation of treatment effects and variances

The history-specific treatment effect estimator (9) is based on the differences in average employment rate outcomes between treatment and comparison units. One notable element of this estimator is that multiple comparison units matched to a single treated unit (due to the oversampling algorithm) are handled as if they were one single comparison unit. The variance for (9) is then composed of the sum of independent single variances of each of the employment rate averages entering (9) for "individual" treated and comparison units. This appendix illustrates the generic calculation of this individual variance, and how this yields variances for (9) and (10).

Within each stratum – defined by pre-treatment labor market history – employment success in the three post-treatment quarters is summarized by the average employment rate $\frac{\sum 1}{3}$. For the unrestricted multinomial model each of the $3^3=27$ possible outcomes is associated with a separate probability. For instance, conditional on the *k*-*th* history the probability to be employed in all subsequent quarters is $p(111|h_k)$, the probability to be employed in the first and unemployed in the following two quarters is $p(122|h_k)$, the probability to be unemployed in the first two and out-of-the-laborforce in the third quarter is $p(220|h_k)$ etc. Let us order the 27 probabilities in the following way

$\frac{\sum 1}{3} = 0$	$\frac{\sum 1}{3} = \frac{1}{3}$	$\frac{\sum 1}{3} = \frac{2}{3}$	$\frac{\sum 1}{3} = 1$
p(000 h _k)=p ₁	p(001 h _k)=p ₉	p(011 h _k)=p ₂₁	p(111 h _k)=p ₂₇
$p(002 h_k)=p_2$	$p(021 h_k) = p_{10}$	$p(211 h_k)=p_{22}$	
$p(020 h_k)=p_3$	$p(201 h_k)=p_{11}$	$p(101 h_k)=p_{23}$	
$p(200 h_k)=p_4$	$p(221 h_k)=p_{12}$	$p(121 h_k)=p_{24}$	
$p(022 h_k)=p_5$	$p(010 h_k)=p_{13}$	$p(110 h_k)=p_{25}$	
$p(202 h_k)=p_6$	$p(012 h_k) = p_{14}$	$p(112 h_k)=p_{26}$	
p(220 h _k)=p ₇	p(210 h _k)=p ₁₅		
$p(222 h_k)=p_8$	$p(212 h_k)=p_{16}$		
	p(100 h _k)=p ₁₇		
	$p(102 h_k) = p_{18}$		
	$p(120 h_k) = p_{19}$		
	$p(122 h_k)=p_{20}$		

where $p_{27} = 1 - \sum_{m=1}^{26} p_m$. Then, for each individual i with history k (suppressing the subscripts h_k for notational convenience)

$$E(\frac{\sum 1}{3}) = E[\frac{1}{3}\sum_{q} \mathbf{1}(Y_{qi} = 1)]$$
(B1)

$$= 0(p_{1} + ... + p_{8}) + \frac{1}{3}(p_{9} + ... + p_{20}) + \frac{2}{3}(p_{21} + ... + p_{26}) + 1p_{27}$$

$$= \frac{1}{3}(p_{9} + ... + p_{20}) + \frac{2}{3}(p_{21} + ... + p_{26}) + (1 - \sum_{m=1}^{26} p_{m})$$

$$= \mathbf{m}$$

and

(B2)

$$Var(\frac{\sum 1}{3}) = (-\mathbf{m})^{2} (p_{1} + ... + p_{8}) + (\frac{1}{3} - \mathbf{m})^{2} (p_{9} + ... + p_{20}) + (\frac{2}{3} - \mathbf{m})^{2} (p_{21} + ... + p_{26}) + (1 - \mathbf{m})^{2} (1 - \sum_{m=1}^{26} p_{m}) = \mathbf{s}^{2}$$

In practice, the p_i are estimated as sample fractions. For the n_h individuals with a common history follows

(B3)
$$E(\frac{1}{n_h}\sum_i \mathbf{m}) = \mathbf{m}_h$$
 and
(B4) $Var(\frac{1}{n_h}\sum_i (\frac{\sum 1}{3})) = \frac{1}{n_h}\mathbf{s}^2 = \mathbf{s}_h^2$

which yields the variance for both elements of the difference in (9). The variance of (9) then results from the sum of the two history-specific variances (B4) for treated and comparison units. Parallel to the derivation of the overall treatment effect (10) from the history-specific effect (9), the variance of (10) is a weighted sum (with squared weights) of the variance of (9).

Table 1. Composition of matched samples

		Tra	ining	Intervent	ion Works
		treated	untreated	treated	untreated
Initial Sample	Observations	121	7177	275	7177
Sample A	Observations	121	6751	275	6757
	age	34.5	33.1	36.3	33.1
	%education ^a	91.7	80.7	64.0	80.7
	%female	56.2	53.0	40.4	53.0
	%married	66.9	65.8	67.6	65.6
Sample B	Observations	114	983	244	1354
	age	34.0	33.0	36.0	34.7
	%education	93.9	98.9	69.3	87.4
	%female	56.1	62.1	40.6	51.9
	%married	65.8	23.2	70.5	77.8
Sample C	Observations	87	111	212	240
L.	[Individuals] ^b		[110]		[211]
	age	33.4	33.8	36.0	35.2
	%education	96.6	97.3	71.2	74.2
	%female	58.6	64.8	42.0	44.6
	%married	67.8	70.3	70.3	70.4

^a Excluding individuals with only primary school attainment or less. ^b Number of observations that the algorithm matched exactly once.

Table 2. Average post-treatment employment rate treatment effect by pretreatment labor market history for comparison sample C – Intervention Works

		treated units		C	comparison unit	S		
job history	Ν	rate ^a	std.err.	Ν	rate	std.err.	effect ^b	std.err.
0000	5	0.333	0.189	6	0.400	0.219	-0.067	0.289
0002	1	0.000	0.000	1	0.667	0.471	-0.667	0.471
1111	16	0.813	0.098	19	0.729	0.111	0.084	0.148
1112	5	0.467	0.202	6	0.167	0.167	0.300	0.262
1122	6	0.222	0.150	6	0.333	0.192	-0.111	0.244
1222	4	0.500	0.250	4	0.833	0.186	-0.333	0.312
2000	1	1.000	0.000	1	0.000	0.000	1.000	0.000
2111	1	1.000	0.000	1	1.000	0.000	0.000	0.000
2211	4	0.167	0.144	4	0.667	0.236	-0.500	0.276
2221	1	0.000	0.000	1	0.333	0.471	-0.333	0.471
2222	168	0.183	0.027	191	0.333	0.036	-0.150	0.045
total ^c	212			240			-0.126	0.040

^a Average employment rate in the three post-treatment quarters. ^b Difference between rates of treated units and matched comparison units.

^c Total effect is the weighted average of the effects for the individual histories using the treated units' sample fractions as weights.

Subdivision by	Categories	treated units	matched comparison units	effect ^a	std.err.
Sample A	-	275	6757	285	.026
Sample B	-	244	1354	291	.031
Sample C:	-	212	240	126	.040
Gender	Men Women	123 89	133 107	236 .026	.051 .062
Date of Program Entry	≤ June 1994 ≥ July 1994	116 96	137 103	135 115	.052 .056
Program Entry & Gender	≤ June 1994 Men ≤ June 1994 Women ≥ July 1994 Men ≥ July 1994 Women	66 50 57 39	73 64 60 43	295 .076 167 038	.069 .079 .073 .089
Labor market history	1111 2222	16 168	19 191	.084 150	.148 .045
Labor market history & Gender	1111 Men 1111 Women 2222 Men 2222 Women	10 6 100 68	12 7 108 83	.117 .028 258 .010	.161 .274 .057 .072

 Table 3. Average post-treatment employment rate treatment effect for subsamples

 – Intervention Works

^a Average employment rate in the three post-treatment quarters.

Subdivision by	Categories	treated units	matched comparison units	effect ^a	std.err.
Sample A	-	121	6751	027	.046
Sample B	-	114	983	048	.049
Sample C:	-	87	111	.138	.059
Gender	Men Women	36 51	39 72	.148 .130	.092 .070
Date of Program Entry	≤ June 1994 ≥ July 1994	38 39	52 59	.212 .080	.088 .064
Program Entry & Gender	≤ June 1994 Men ≤ June 1994 Women ≥ July 1994 Men ≥ July 1994 Women	15 23 21 28	17 35 22 37	.056 .313 .214 020	.156 .104 .094 .086
Labor market history	1111 2222	24 32	34 43	.071 077	.115 .103
Labor market history & Gender	1111 Men 1111 Women 2222 Men 2222 Women	11 13 11 21	12 22 12 31	.045 .092 046 .093	.194 .129 .192 .116

Table 4. Average post-treatment employment rate treatment effect for subsamples– Training

^a Average employment rate in the three post-treatment quarters.

Treatment	Weights	Effect ^a	Std.Err.	Interpretation
Intervention Works	Intervention Works	126	.040	Factual IW treatment effect
Intervention Works	Training	048	.064	Counterfactual IW treatment effect
Training	Training	.138	.059	Factual Training treatment effect
Training	Intervention Works	.089	.083	Counterfactual Training treatment effect
Intervention Works – Training	Intervention Works	218	.093	Differential treatment effect Intervention Works vs. Training
Training – Intervention Works	Training	.185	.087	Differential treatment effect Training vs. Intervention Works

Table 5. Counterfactual treatment effects for samples C

^a Average employment rate in the three post-treatment quarters.





1st control for #2

_____C _____

Figure 2. Distribution of age – Intervention Works





Kernel density estimates of the relevant variable for treated and comparison units by STATA using an Epanechnikov kernel and total bandwidth of (.5). Density estimates are not bound, their purpose is for illustration only.

Figure 3. Distribution of region – Intervention Works





Region = 49 voivodships.

Figure 4. Distribution of pre-treatment labor market history by sample – Intervention Works





Sample B



The 3⁴ possible labor force status sequences are classified into 11 categories (see text and Appendix A).

Figure 5. Distribution of pre-treatment labor market history by sample – Training





Sample B



The 3⁴ possible labor force status sequences are classified into 11 categories (see text and Appendix A).

Figure 6. Distribution of estimated propensity score by sample – Intervention Works









Kernel density estimates of the propensity score for treated and comparison units by STATA using an Epanechnikov kernel and total bandwidth of (.02). Density estimates are not bound, their purpose is for illustration only. Y-axis denotes percentages.

Figure 7. Distribution of estimated propensity score by sample – Training









Kernel density estimates of the propensity score for treated and comparison units by STATA using an Epanechnikov kernel and total bandwidth of (.02). Density estimates are not bound, their purpose is for illustration only. Y-axis denotes percentages.

Figure 8. Distribution of post-treatment labor market sequence by sample – Intervention Works









Sample C



The 3³ possible labor force status sequences are classified into 9 categories (see text and Appendix A).

Figure 9. Distribution of post-treatment labor market sequence by sample – Training













The 3³ possible labor force status sequences are classified into 9 categories (see text and Appendix A).

During 3 months BEFORE treatment:





During 3 months AFTER treatment:





During 9 months AFTER treatment:



The upper panel indicates benefit receipt (="yes") during at least two of the last three months preceding treatment. The middle panel indicates benefit receipt during at least two of the first three months succeeding treatment. The bottom panel indicates benefit receipt during at least two of the three months in <u>each</u> of the three quarters succeeding treatment.

During 3 months BEFORE treatment:





During 3 months AFTER treatment:



During 9 months AFTER treatment:



The upper panel indicates benefit receipt (="yes") during at least two of the last three months preceding treatment. The middle panel indicates benefit receipt during at least two of the first three months succeeding treatment. The bottom panel indicates benefit receipt during at least two of the three months in <u>each</u> of the three quarters succeeding treatment.

IZA Discussion Papers

No.	Author(s)	Title	Area	Date
280	P. Apps R. Rees	Household Saving and Full Consumption over the Life Cycle	7	04/01
281	G. Saint-Paul	Information Technology and the Knowledge Elites	5	04/01
282	J. Albrecht A. Björklund S. Vroman	Is There a Glass Ceiling in Sweden?	5	04/01
283	M. Hagedorn A. Kaul V. Reinthaler	Welfare Analysis in a Schumpeterian Growth Model with Capital	7	04/01
284	H. Rapoport A. Weiss	The Optimal Size for a Minority	1	04/01
285	J. Jerger C. Pohnke A. Spermann	Gut betreut in den Arbeitsmarkt? Eine mikroökonometrische Evaluation der Mannheimer Arbeitsvermittlungsagentur	5	04/01
286	M. Fertig C. M. Schmidt	First- and Second-Generation Migrants in Germany – What Do We Know and What Do People Think	1	04/01
287	P. Guggenberger A. Kaul M. Kolmar	Efficiency Properties of Labor Taxation in a Spatial Model of Restricted Labor Mobility	3	04/01
288	D. A. Cobb-Clark	Getting Ahead: The Determinants of and Payoffs to Internal Promotion for Young U.S. Men and Women	5	04/01
289	L. Cameron D. A. Cobb-Clark	Old-Age Support in Developing Countries: Labor Supply, Intergenerational Transfers and Living Arrangements	3	04/01
290	D. A. Cobb-Clark M. D. Connolly C. Worswick	The Job Search and Education Investments of Immigrant Families	1	04/01
291	R. T. Riphahn	Cohort Effects in the Educational Attainment of Second Generation Immigrants in Germany: An Analysis of Census Data	1	05/01

292	E. Wasmer	Between-group Competition in the Labor Market and the Rising Returns to Skill: US and France 1964-2000	5	05/01
293	D. Cobb-Clark T. F. Crossley	Gender, Comparative Advantage and Labor Market Activity in Immigrant Families	1	05/01
294	Š. Jurajda	Estimating the Effect of Unemployment Insurance Compensation on the Labor Market Histories of Displaced Workers	3	05/01
295	F. Duffy P. P. Walsh	Individual Pay and Outside Options: Evidence from the Polish Labour Force Survey	4	05/01
296	H. S. Nielsen M. Rosholm N. Smith L. Husted	Intergenerational Transmissions and the School- to-Work Transition of 2 nd Generation Immigrants	1	05/01
297	J. C. van Ours J. Veenman	The Educational Attainment of Second Generation Immigrants in The Netherlands	1	05/01
298	P. Telhado Pereira P. Silva Martins	Returns to Education and Wage Equations	5	06/01
299	G. Brunello C. Lucifora R. Winter-Ebmer	The Wage Expectations of European College Students	5	06/01
300	A. Stutzer R. Lalive	The Role of Social Work Norms in Job Searching and Subjective Well-Being	5	06/01
301	J. R. Frick G. G. Wagner	Economic and Social Perspectives of Immigrant Children in Germany	1	06/01
302	G. S. Epstein A. Weiss	A Theory of Immigration Amnesties	1	06/01
303	G. A. Pfann B. F. Blumberg	Social Capital and the Uncertainty Reduction of Self-Employment	5	06/01
304	P. Cahuc E. Wasmer	Labour Market Efficiency, Wages and Employ- ment when Search Frictions Interact with Intra- firm Bargaining	2	06/01
305	H. Bonin	Fiskalische Effekte der Zuwanderung nach Deutschland: Eine Generationenbilanz	1	06/01

306	H. Bonin G. Abío E. Berenguer J. Gil C. Patxot	Is the Deficit under Control? A Generational Accounting Perspective on Fiscal Policy and Labour Market Trends in Spain	2	06/01
307	G. A. Pfann	Downsizing	1/5	06/01
308	G. A. Pfann D. S. Hamermesh	Two-Sided Learning, Labor Turnover and Worker Displacement	1	06/01
309	G. Brunello	On the Complementarity between Education and Training in Europe	5	06/01
310	U. Sunde	Human Capital Accumulation, Education and Earnings Inequality	5	06/01
311	G. Brunello	Unemployment, Education and Earnings Growth	3	06/01
312	C. Furnée M. Kemler G. A. Pfann	The Value of Pain Relief	5	06/01
313	A. Ferrer-i-Carbonell B. M.S. van Praag	The Subjective Costs of Health Losses due to Chronic Diseases: An Alternative Model for Monetary Appraisal	7	06/01
314	B. M.S. van Praag A. Ferrer-i-Carbonell	Age-Differentiated QALY Losses	7	06/01
315	W. H. J. Hassink R. Schettkat	On Price-Setting for Identical Products in Markets without Formal Trade Barriers	7	06/01
316	M. Frondel C. M. Schmidt	Rejecting Capital-Skill Complementarity at all Costs	5	06/01
317	R. Winkelmann	Health Care Reform and the Number of Doctor Visits – An Econometric Analysis	7	06/01
318	M. Pannenberg G. G. Wagner	Overtime Work, Overtime Compensation and the Distribution of Economic Well-Being: Evidence for West Germany and Great Britain	1	06/01
319	R. Euwals R. Winkelmann	Why do Firms Train? Empirical Evidence on the First Labour Market Outcomes of Graduated Apprentices	1	06/01

320	R. Fahr U. Sunde	Strategic Hiring Behavior in Empirical Matching Functions	1	06/01
321	P. Telhado Pereira P. Silva Martins	Is there a Return – Risk Link in Education?	5	07/01
322	O. Hübler U. Jirjahn	Works Councils and Collective Bargaining in Germany: The Impact on Productivity and Wages	1	07/01
323	A. Frederiksen E. K. Graversen N. Smith	Overtime Work, Dual Job Holding and Taxation	1	07/01
324	M. Pflüger	Trade, Technology and Labour Markets: Empirical Controversies in the Light of the Jones Model	2	07/01
325	R. A. Hart J. R. Malley U. Woitek	Real Wages and the Cycle: The View from the Frequency Domain	1	07/01
326	J. S. Earle Á. Telegdy	Privatization and Productivity in Romanian Industry: Evidence from a Comprehensive Enterprise Panel	4	07/01
327	H. Gersbach A. Schmutzler	A Product Market Theory of Training and Turnover in Firms	5	07/01
328	F. Breyer	Why Funding is not a Solution to the "Social Security Crisis"	3	07/01
329	X. Gong A. van Soest	Wage Differentials and Mobility in the Urban Labor Market: A Panel Data Analysis for Mexico	1	07/01
330	D. N. Margolis K. G. Salvanes	Do Firms Really Share Rents with Their Workers?	5	07/01
331	R. Winkelmann	Why Do Firms Recruit Internationally? Results from the IZA International Employer Survey 2000	5	07/01
332	M. Rosholm	An Analysis of the Processes of Labour Market Exclusion and (Re-) Inclusion	3	07/01
333	W. Arulampalam R. A. Naylor J. P. Smith	A Hazard Model of the Probability of Medical School Dropout in the United Kingdom	5	07/01

334	P. A. Puhani	Wage Rigidities in Western Germany? Microeconometric Evidence from the 1990s	1	07/01
335	R. Fahr U. Sunde	Disaggregate Matching Functions	1	07/01
336	F. Lima P. Telhado Pereira	Careers and Wage Growth within Large Firms	5	07/01
337	F. Büchel M. Pollmann-Schult	Overeducation and Skill Endowments: The Role of School Achievement and Vocational Training Quality	5	08/01
338	C. Bell H. Gersbach	Child Labor and the Education of a Society	5	08/01
339	A. Ibourk B. Maillard S. Perelman H. R. Sneessens	The Matching Efficiency of Regional Labour Markets: A Stochastic Production Frontier Estimation, France 1990-1995	1	08/01
340	X. Wauthy Y. Zenou	How Does Imperfect Competition in the Labor Market Affect Unemployment Policies?	3	08/01
341	S. Kohns	Testing for Asymmetry in British, German and US Unemployment Data	1	08/01
342	W. Schnedler	The Virtue of Being Underestimated: A Note on Discriminatory Contracts in Hidden Information Models	5	08/01
343	H. Bonin	Will it Last? An Assessment of the 2001 German Pension Reform	3	08/01
344	E. Plug P. Berkhout	Effects of Sexual Preferences on Earnings in the Netherlands	5	08/01
345	J. Hampe M. Steininger	Survival, Growth, and Interfirm Collaboration of Start-Up Companies in High-Technology Industries: A Case Study of Upper Bavaria	5	08/01
346	L. Locher	The Determination of a Migration Wave Using Ethnicity and Community Ties	1	08/01
347	M. Lofstrom F. D. Bean	Labor Market Conditions and Post-Reform Declines in Welfare Receipt Among Immigrants	3	08/01
348	S. Neuman A. Ziderman	Can Vocational Education Improve the Wages of Minorities and Disadvantaged Groups? The Case of Israel	5	08/01

349	J. T. Addison P. Portugal	Job Search Methods and Outcomes	1	08/01
350	J. T. Addison P. Portugal	Unemployment Duration: Competing and Defective Risks	1	08/01
351	J. D. Brown J. S. Earle	Gross Job Flows in Russian Industry Before and After Reforms: Has Destruction Become More Creative?	4	08/01
352	J. T. Addison J. S. Heywood X. Wei	Unions and Plant Closings in Britain: New Evidence from the 1990/98 WERS	1	08/01
353	T. Bauer S. Bender	Flexible Work Systems and the Structure of Wages: Evidence from Matched Employer- Employee Data	5	08/01
354	J. Kluve	On the Role of Counterfactuals in Inferring Causal Effects of Treatments	6	09/01
355	J. Kluve H. Lehmann C. M. Schmidt	Disentangling Treatment Effects of Polish Active Labor Market Policies: Evidence from Matched Samples	4/6	09/01